Author’s response to reviews

Title: Fetus in Fetu: Case report and review of literature

Authors:

Nisreen M Khalifa (nisreen_khalifa@hotmail.com)
Doaa W Maximous (doaadavid@hotmail.com)
Alaa A Abd-Elsayed (alaaawny@hotmail.com)

Version: 2 Date: 26 September 2007

Author’s response to reviews:

We edited the manuscript so it has a separate title page.
removed references in the abstract.
We included an Introduction
We changed the section marked Case Report to Case Presentation.
We removed the figures from the manuscript (they should appear at the end as separate pages)
Figure legends appear before the references.
We added a Conclusion subheading.
We removed some references and the total number now is 10 reference.
The manuscript contains only 3 figures.
We included Competing interests declaration
We included an Authors contribution list
We included an Acknowledgement list.